Author's response to reviews

Title: Latrine coverage and associated factors among rural communities in Bahir Dar Zuria District, West Gojjam Zone, Amhara Region, Ethiopia. 2012

Authors:

Worku A Yalew (workuawo@yahoo.com)
2. Semahegn M Bayabil (semahegnmuche@yahoo.com)

Version: 2 Date: 2 September 2012

Author's response to reviews:

Dears, after reviewing the document that we upload previously we understood that the abstract section was messing. Therefore we kindly request for reconsidering it.

With kind regards
Work Awoke Yalew